Author's response to reviews

Title: Monitoring the health of gender minority populations: Cognitive testing of natal sex and gender identity survey items in a U.S. national cohort of young adults

Authors:

Sari L Reisner (sreisner@hsph.harvard.edu)
Kerith J Conron (kconron@fenwayhealth.org)
Laura Anatale Tardiff (laura.anatale.tardiff@channing.harvard.edu)
Stephanie Jarvi (stephanie.jarvi@gmail.com)
Allegra R Gordon (argordon@mail.harvard.edu)
S Bryn Austin (Bryn.Austin@childrens.harvard.edu)

Version: 2 Date: 13 May 2014

Author's response to reviews:

Uploading manuscript with line numbers, author emails, and statement of no competing interests.